Mauna Loa Solar Observatory Observer's Log _____ Sat May 27 16:28:32 GMT 2017 Year: 17 Doy: 147 Observer: stueben WEATHER COMMENT: stueben: Sat May 27 16:29:28 GMT 2017 Temp: 38.2f, Humidity: 19%, Pressure: 28.758in, Wind: Southwest 5 mph, Skies: Clear. end Sat May 27 16:46:59 GMT 2017 CoMP occulter has been re-centered Sat May 27 16:47:08 GMT 2017 COMP Start Patrol on cookbook: synoptic-00010.cbk Sat May 27 19:28:53 GMT 2017 COMP End Patrol Sat May 27 19:28:54 GMT 2017 COMP Start Patrol on cookbook: waves-00001.cbk Sat May 27 20:36:08 GMT 2017 CoMP Paused for clouds Sat May 27 20:40:43 GMT 2017 CoMP Restarted from pause Sat May 27 20:43:08 GMT 2017 CoMP Paused for clouds Sat May 27 21:24:07 GMT 2017 CoMP Restarted from pause Sat May 27 21:24:27 GMT 2017 COMP End Patrol WEATHER COMMENT: stueben: Sat May 27 23:22:32 GMT 2017 Dense clouds and blowing fog, dome closed. end **KCOR PROBLEM COMMENT BY stueben** : Sat May 27 23:34:19 GMT 2017 The glue which Ben applied to the front dusk cover yesterday seemed well cured so after the clouds came up I reattached th e dust cover and mounted the 01. All went well, though slowly, since its been a while since I've worked with that assemble y. end WEATHER COMMENT: stueben: Sun May 28 01:27:09 GMT 2017 Still cloudy. end GENERAL COMMENT BY stueben: Sun May 28 01:28:26 GMT 2017 Nice morning with fair seeing but clouds came up by about 10:30. end